Author's response to reviews

Title: Kidney Function and Comorbid Disorders in Patients with Kidney Stone Disease

Authors:

Vaka K Sigurjonsdottir (vakaks@gmail.com)
Hrafnhildur L Runolfsdottir (hilda_linnet@hotmail.com)
Olafur S Indridason (olasi@lsh.is)
Runolfur Palsson (runolfur@lsh.is)
Vidar O Edvardsson (vidare@lsh.is)

Version: 2 Date: 6 February 2015

Author's response to reviews:

The requested changes have been made to the manuscript.

1. Line- and page numbering are now included. All page brakes have been deleted.
2. Title page – E-mail addresses of all authors have been added to the Title Page.
3. Consent – The following statement has been added to the Methods section: “A written informed consent was obtained from all participants in the study.” (No children were included, all participants had reached at least 18 years of age at the time of participation.)